Author’s response to reviews

Title: SMS SOS: a randomized controlled trial to reduce self-harm and suicide attempts using SMS text messaging

Authors:

Garry Stevens (g.stevens@westernsydney.edu.au)
Trent Hammond (trent.hammond@sydney.edu.au)
Suzanne Brownhill (suzanne.brownhill@health.nsw.gov.au)
Manish Anand (manish.anand@health.nsw.gov.au)
Anabel De la Riva (anabel.delariva@health.nsw.gov.au)
Jean Hawkins (jean.hawkins@health.nsw.gov.au)
Tristan Chapman (tristan.chapman@health.nsw.gov.au)
Richard Baldacchino (richard.baldacchino@health.nsw.gov.au)
Jo-Anne Micallef (joanne.micallef@health.nsw.gov.au)
Jagadeesh Andepalli (jagadeesh.andepalli@health.nsw.gov.au)
Anita Kotak (anita.kotak@health.nsw.gov.au)
Naren Gunja (Naren.Gunja@health.nsw.gov.au)
Andrew Page (a.page@westernsydney.edu.au)
Grahame Gould (grahame@gouldpsychology.com.au)
Christopher Ryan (christopher.ryan@sydney.edu.au)
Ian Whyte (ian.whyte@newcastle.edu.au)
Gregory Carter (gregory.carter@newcastle.edu.au)
Alison Jones (alisonj@uow.edu.au)

Version: 2 Date: 04 Apr 2019

Author’s response to reviews:
April 4 2019

Dear Dr Ragen,

RE: Your submission to BMC Psychiatry - BPSY-D-18-00726R1

Regarding the email we received from BMC Psychiatry on April 3 2019, please find attached a revised study protocol manuscript and figures for the submission: BPSY-D-18-00726 “SMS SOS: a randomized controlled trial to reduce self-harm and suicide attempts using SMS text messaging.”

Commensurate with BMC Psychiatry’s Submission Guidelines, several small revisions have been made to the manuscript to improve transparency, readability, and understanding. As such, the provision of specific line numbers for these revisions would not be prudent.

Kindly find attached the newest version of our manuscript for review.

We look forward to a positive outcome regarding the publication of our manuscript in BMC Psychiatry.
Yours Sincerely,

Dr. Garry Stevens

Project Manager, SMS SOS Study

Mr Trent Hammond

Corresponding Author, SMS SOS Study